Author's response to reviews

Title: Cohort studies in chronic diseases using routinely collected databases when a prescription is used as surrogate outcome

Authors:

Sara Lodi (sal@ctu.mrc.ac.uk)
James Carpenter (james.carpenter@lshtm.ac.uk)
Peter Egger (peter.j.egger@gsk.com)
Stephen Evans (stephen.evans@lshtm.ac.uk)

Version: 2 Date: 14 September 2010

Author's response to reviews:

Dear Editor,

Please find attached the amended version of the manuscript in response to your request for more information on the database used for the analysis and the revision of the conflict of interest section.

All the best